Author's response to reviews

Title: What all students in healthcare training programs should learn to increase health equity: Perspectives on postcolonialism and health

Authors:

Allana SW Beavis (allanabeavis@gmail.com)
Ala Hojjati (ala.hojjati@mail.utoronto.ca)
Aly Kassam (kassam057@gmail.com)
Daniel Choudhury (daniel.choudhury@mail.utoronto.ca)
Michelle Fraser (mmhfraser@hotmail.ca)
Renee Masching (reneem@caan.ca)
Stephanie A Nixon (stephanie.nixon@utoronto.ca)

Version: 3
Date: 10 February 2015

Author's response to reviews:

Dear Editors,

Thank you kindly for your feedback on our submission.

The following changes have been made to main body of the manuscript as requested:

1. In the "Methods" section, a statement has now been included regarding informed consent. It reads as follows: "Informed consent was obtained verbally from participants."

Sincerely,
Allana Beavis
Corresponding Author